Author's response to reviews

Title: Influence of antioxidant (l- ascorbic acid) on tolbutamide induced hypoglycaemia/antihyperglycaemia in normal and diabetic rats

Authors:

Satyanarayana Sreemantula (nandinisai@hotmail.com)
Eswar K Kilari (ekilari@rediffmail.com)
VishnuVardhan A (vishani_pharma@rediffmail.com)
Rajasekhar Jaladi (rajasekhar_27@hotmail.com)

Version: 5 Date: 9 February 2005

Author's response to reviews: see over
Dear Dr. Alison Z Shaw,

Thank you very much for accepting our paper (MS: 136550733385237) for publication in BMC Endocrine Disorders. I am herewith submitting the modified final version of our manuscript as per your journal format. The following changes were made as per your suggestion.

- ‘Conclusion’ to ‘Conclusions’
- ‘supplementation’ to ‘Supplementation’
- Removed the key words section from the manuscript.
- Renamed the ‘Materials & methods’ into ‘Methods’
- Table-I & II to Table-1 & 2.
- Included the authors’ contributions in the manuscript.
- References were formatted as per the journal’s style.
- Figures were removed from the manuscript and were uploaded as separate files.
- Figures were cropped as per the journal’s requirement.

Hope the changes made will be sufficient for publication and I appreciate your early response in this regard.

Thanking you
Sincerely yours
Satyanarayana Sreemantula